Author's response to reviews

Title: A long-term memory of HIF induction in response to chronic mild decreased oxygen after oxygen normalization

Authors:

Chandrashekhar D Kamat (shekhar-kamat@ouhsc.edu)
Jessica E Thorpe (jessica-thorpe@ouhsc.edu)
Satyendra S Shenoy (satyendra-shenoy@ouhsc.edu)
Antonio Ceriello (antonio.ceriello@warwick.ac.uk)
Dixy E Green (dixy-green@ouhsc.edu)
Linda A Warnke (linda-warnke@ouhsc.edu)
Michael A Ihnat (michael-ihn@ouhsc.edu)

Version: 4 Date: 18 January 2007

Author's response to reviews:

The title was made sentence case
Mr. Shenoy's middle initial was added to the title page
A statement was added to authors contributions "All authors read and approved the final manuscript."